Author's response to reviews

Title: The first report of RPSA polymorphisms, also called 37/67 kDa LRP/LR gene, in sporadic Creutzfeldt-Jakob disease (CJD)

Authors:

Jisuk Yun (jigel8@hanmail.net)
Hyoung-Tae Jin (htjin@hallym.ac.kr)
Yun-Jung Lee (jiung0301@hallym.ac.kr)
Eun-Kyoung Choi (ekchoi@hallym.ac.kr)
Richard I. Carp (richard.carp@opwdd.ny.gov)
Byung-Hoon Jeong (bhjeong@hallym.ac.kr)
Yong-Sun Kim (yskim@hallym.ac.kr)

Version: 5 Date: 11 August 2011

Author's response to reviews: see over
Dear The BMC Series Editorial Production Team

We have revised our manuscript in accordance with the suggestions made by the editorial production team and are therefore submitting the revised manuscript.

In relation to the comments of the editor production team, our corrections and revisions are as follows:

**Major revisions**

Q) Abstract: this abstract does not have the correct structure, please use these headings: Background, Methods, Results, Conclusions. Please remember to update the abstract in the submission system.

Response: According to the editorial team’s suggestion, we had corrected the structure of Abstract.

Q) Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

Response: We removed all highlighting and tracking from our manuscript.

**Minor revisions**

Q) Spacing: please try to reduce blank space by reducing the spacing between sections. For example, Competing interests, Authors' Contributions, and Acknowledgements can all feature on the same page.

Response: According to the editorial team’s suggestion, we had reduced the blank space.

Q) Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes
Response: As suggested by the editorial team’s suggestion, we had corrected the typographical errors.

Very sincerely yours,

Yong-Sun Kim, M.D., Ph.D.

Ilsong Institute of Life Science,
Hallym University, Ilsong Building,
1605-4 Gwanyang-dong, Dongan-gu, Anyang,
Gyeonggi-do 431-060, South Korea
TEL : 82-31-380-1986, FAX : 82-31-388-3427
E-mail : yskim@hallym.ac.kr